Dear Editor in Chief of “World Journal of Gastroenterology”

Dear Reviewer,

We thank you very much for consideration of our manuscript “Transition after pediatric liver transplantation - perceptions of adults, adolescents and parents” for publication in World Journal of Gastroenterology.

We are grateful for the constructive feedback and comments on our manuscript. We believe that we could further improve it due to the feedback and comments.

We are in complete agreement with reviewer 1560464 in point 1).
In point 2) the reviewer very nicely summarizes our conclusions and underlines the importance of our study. In point 3) the reviewer asked for further improvement, which we hopefully could realize by changes in the manuscript and language improvement. Collaboration with multiple transplant centers is an important aspect for further studies. Also to compare patient’s perceptions between different centers. However for this study we aimed to evaluate perceptions of our patients, therefore it was a single center study.

We thank reviewer 02861333 for accepting our manuscript for publication.

Sincerely

Norman Junge, MD